\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{at192a.txt}{\rawdatadir/at192a.txt}~Abid (1991)~\cite{s_abid91}
\W\label{at192a}
\W\end{table}
\W\end{center}
\T\begin{longtable}{ccc}
\T\caption{at192a~\protect\cite[Abid (1991)]{s_abid91}\label{at192a}}\\
\T\input{\datadir/at192a.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & cell size & Fuel: & CH4 \\
Sub-Category: & width & Oxidizer: & O2 \\
Initial Pressure: & 6.5-12 kPa & Diluent: &  \\
Initial Temperature: & 293 K & Equivalence Ratio: & 1 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{ja6a.txt}{\rawdatadir/ja6a.txt}~Akbar (1997)~\cite{r_akbar97}
\W\label{ja6a}
\W\end{table}
\W\end{center}
\T\begin{longtable}{cc}
\T\caption{ja6a~\protect\cite[Akbar (1997)]{r_akbar97}\label{ja6a}}\\
\T\input{\datadir/ja6a.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & cell size & Fuel: & CH4 \\
Sub-Category: & width & Oxidizer: & O2 \\
Initial Pressure: & 70-72 kPa & Diluent: & N2 \\
Initial Temperature: & 293 K & Equivalence Ratio: & 1 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{ja6b.txt}{\rawdatadir/ja6b.txt}~Akbar (1997)~\cite{r_akbar97}
\W\label{ja6b}
\W\end{table}
\W\end{center}
\T\begin{longtable}{cc}
\T\caption{ja6b~\protect\cite[Akbar (1997)]{r_akbar97}\label{ja6b}}\\
\T\input{\datadir/ja6b.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & cell size & Fuel: & CH4 \\
Sub-Category: & width & Oxidizer: & O2 \\
Initial Pressure: & 102 kPa & Diluent: & N2 \\
Initial Temperature: & 293 K & Equivalence Ratio: & 1 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{ja7a.txt}{\rawdatadir/ja7a.txt}~Akbar (1997)~\cite{r_akbar97}
\W\label{ja7a}
\W\end{table}
\W\end{center}
\T\begin{longtable}{cc}
\T\caption{ja7a~\protect\cite[Akbar (1997)]{r_akbar97}\label{ja7a}}\\
\T\input{\datadir/ja7a.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & cell size & Fuel: & CH4 \\
Sub-Category: & width & Oxidizer: & N2O \\
Initial Pressure: & 57-72 kPa & Diluent: & N2 \\
Initial Temperature: & 293 K & Equivalence Ratio: & 1 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{ja7b.txt}{\rawdatadir/ja7b.txt}~Akbar (1997)~\cite{r_akbar97}
\W\label{ja7b}
\W\end{table}
\W\end{center}
\T\begin{longtable}{cc}
\T\caption{ja7b~\protect\cite[Akbar (1997)]{r_akbar97}\label{ja7b}}\\
\T\input{\datadir/ja7b.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & cell size & Fuel: & CH4 \\
Sub-Category: & width & Oxidizer: & N2O \\
Initial Pressure: & 77-87 kPa & Diluent: & N2 \\
Initial Temperature: & 293 K & Equivalence Ratio: & 1 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{ja7c.txt}{\rawdatadir/ja7c.txt}~Akbar (1997)~\cite{r_akbar97}
\W\label{ja7c}
\W\end{table}
\W\end{center}
\T\begin{longtable}{cc}
\T\caption{ja7c~\protect\cite[Akbar (1997)]{r_akbar97}\label{ja7c}}\\
\T\input{\datadir/ja7c.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & cell size & Fuel: & CH4 \\
Sub-Category: & width & Oxidizer: & N2O \\
Initial Pressure: & 92-97 kPa & Diluent: & N2 \\
Initial Temperature: & 293 K & Equivalence Ratio: & 1 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{ja7d.txt}{\rawdatadir/ja7d.txt}~Akbar (1997)~\cite{r_akbar97}
\W\label{ja7d}
\W\end{table}
\W\end{center}
\T\begin{longtable}{cc}
\T\caption{ja7d~\protect\cite[Akbar (1997)]{r_akbar97}\label{ja7d}}\\
\T\input{\datadir/ja7d.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & cell size & Fuel: & CH4 \\
Sub-Category: & width & Oxidizer: & N2O \\
Initial Pressure: & 102 kPa & Diluent: & N2 \\
Initial Temperature: & 293 K & Equivalence Ratio: & 1 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{ja7e.txt}{\rawdatadir/ja7e.txt}~Akbar (1997)~\cite{r_akbar97}
\W\label{ja7e}
\W\end{table}
\W\end{center}
\T\begin{longtable}{cc}
\T\caption{ja7e~\protect\cite[Akbar (1997)]{r_akbar97}\label{ja7e}}\\
\T\input{\datadir/ja7e.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & cell size & Fuel: & CH4 \\
Sub-Category: & width & Oxidizer: & N2O \\
Initial Pressure: & 86-97 kPa & Diluent: & Air \\
Initial Temperature: & 293 K & Equivalence Ratio: & 0.5-0.7 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{at193b.txt}{\rawdatadir/at193b.txt}~Aminallah (1993)~\cite{m_amin93}
\W\label{at193b}
\W\end{table}
\W\end{center}
\T\begin{longtable}{cc}
\T\caption{at193b~\protect\cite[Aminallah (1993)]{m_amin93}\label{at193b}}\\
\T\input{\datadir/at193b.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & cell size & Fuel: & CH4 \\
Sub-Category: & width & Oxidizer: & O2 \\
Initial Pressure: & 120 kPa & Diluent: &  \\
Initial Temperature: & 293 K & Equivalence Ratio: & 0.7-1.3 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{at176a.txt}{\rawdatadir/at176a.txt}~Beeson (1991)~\cite{h_bees91}
\W\label{at176a}
\W\end{table}
\W\end{center}
\T\begin{longtable}{ccc}
\T\caption{at176a~\protect\cite[Beeson (1991)]{h_bees91}\label{at176a}}\\
\T\input{\datadir/at176a.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & cell size & Fuel: & CH4 \\
Sub-Category: & width & Oxidizer: & Air \\
Initial Pressure: & 101.3 kPa & Diluent: &  \\
Initial Temperature: & 293 K & Equivalence Ratio: & 1 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{at13a.txt}{\rawdatadir/at13a.txt}~Knystautas (1984)~\cite{r_knys84}
\W\label{at13a}
\W\end{table}
\W\end{center}
\T\begin{longtable}{cccc}
\T\caption{at13a~\protect\cite[Knystautas (1984)]{r_knys84}\label{at13a}}\\
\T\input{\datadir/at13a.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & cell size & Fuel: & CH4 \\
Sub-Category: & width & Oxidizer: & Air \\
Initial Pressure: & 101.3 kPa & Diluent: &  \\
Initial Temperature: & 293 K & Equivalence Ratio: & 1 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{at73.txt}{\rawdatadir/at73.txt}~Knystautas (1982)~\cite{r_knys82}
\W\label{at73}
\W\end{table}
\W\end{center}
\T\begin{longtable}{ccc}
\T\caption{at73~\protect\cite[Knystautas (1982)]{r_knys82}\label{at73}}\\
\T\input{\datadir/at73.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & cell size & Fuel: & CH4 \\
Sub-Category: & width & Oxidizer: & O2 \\
Initial Pressure: & 8-13.3 kPa & Diluent: &  \\
Initial Temperature: & 293 K & Equivalence Ratio: & 1 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{at199a.txt}{\rawdatadir/at199a.txt}~Laberge (1993)~\cite{s_labe93}
\W\label{at199a}
\W\end{table}
\W\end{center}
\T\begin{longtable}{ccc}
\T\caption{at199a~\protect\cite[Laberge (1993)]{s_labe93}\label{at199a}}\\
\T\input{\datadir/at199a.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & cell size & Fuel: & CH4 \\
Sub-Category: & width & Oxidizer: & O2 \\
Initial Pressure: & 8-26.7 kPa & Diluent: &  \\
Initial Temperature: & 293 K & Equivalence Ratio: & 1 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{at57d.txt}{\rawdatadir/at57d.txt}~Manzhalei (1974)~\cite{v_manz74}
\W\label{at57d}
\W\end{table}
\W\end{center}
\T\begin{longtable}{ccc}
\T\caption{at57d~\protect\cite[Manzhalei (1974)]{v_manz74}\label{at57d}}\\
\T\input{\datadir/at57d.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & cell size & Fuel: & CH4 \\
Sub-Category: & width & Oxidizer: & O2 \\
Initial Pressure: & 30.4-912 kPa & Diluent: &  \\
Initial Temperature: & 293 K & Equivalence Ratio: & 1 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{at157a.txt}{\rawdatadir/at157a.txt}~Moen (1984)~\cite{i_moen84}
\W\label{at157a}
\W\end{table}
\W\end{center}
\T\begin{longtable}{ccc}
\T\caption{at157a~\protect\cite[Moen (1984)]{i_moen84}\label{at157a}}\\
\T\input{\datadir/at157a.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & cell size & Fuel: & CH4 \\
Sub-Category: & width & Oxidizer: & Air \\
Initial Pressure: & 92.5 kPa & Diluent: &  \\
Initial Temperature: & 293 K & Equivalence Ratio: & 1 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{at184a.txt}{\rawdatadir/at184a.txt}~Pedley (1988)~\cite{m_pedl88}
\W\label{at184a}
\W\end{table}
\W\end{center}
\T\begin{longtable}{ccc}
\T\caption{at184a~\protect\cite[Pedley (1988)]{m_pedl88}\label{at184a}}\\
\T\input{\datadir/at184a.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & cell size & Fuel: & CH4 \\
Sub-Category: & width & Oxidizer: & O2 \\
Initial Pressure: & 7-13 kPa & Diluent: &  \\
Initial Temperature: & 293 K & Equivalence Ratio: & 1 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{at128a.txt}{\rawdatadir/at128a.txt}~Strehlow (1969)~\cite{r_stre69b}
\W\label{at128a}
\W\end{table}
\W\end{center}
\T\begin{longtable}{ccc}
\T\caption{at128a~\protect\cite[Strehlow (1969)]{r_stre69b}\label{at128a}}\\
\T\input{\datadir/at128a.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & cell size & Fuel: & CH4 \\
Sub-Category: & width & Oxidizer: & O2 \\
Initial Pressure: & 15.2-50.7 kPa & Diluent: & Ar \\
Initial Temperature: & 293 K & Equivalence Ratio: & 1 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{at171a.txt}{\rawdatadir/at171a.txt}~Tieszen (1991)~\cite{s_ties91}
\W\label{at171a}
\W\end{table}
\W\end{center}
\T\begin{longtable}{cc}
\T\caption{at171a~\protect\cite[Tieszen (1991)]{s_ties91}\label{at171a}}\\
\T\input{\datadir/at171a.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & cell size & Fuel: & CH4 \\
Sub-Category: & width & Oxidizer: & Air \\
Initial Pressure: & 101.3 kPa & Diluent: &  \\
Initial Temperature: & 298-373 K & Equivalence Ratio: & 1 \\
\end{tabular}
\end{center}

